\begin{thebibliography}{10}

\bibitem{TIMES:03}
T.~Amnell, E.~Fersman, L.~Mokrushin, P.~Pettersson, and W.~Yi.
\newblock {TIMES: A Tool for Schedulability Analysis and Code Generation of
  Real-Time Systems}.
\newblock In K.~G. Larsen and P.~Niebert, editors, {\em Formal Modeling and
  Analysis of Timed Systems: First International Workshop, {FORMATS} 2003,
  Marseille, France, September 6-7, 2003. Revised Papers}, volume 2791 of {\em
  Lecture Notes in Computer Science}, pages 60--72. Springer, 2003.

\bibitem{Schmidt:04z}
K.~Balasubramanian, J.~Balasubramanian, J.~Parsons, A.~Gokhale, and D.~C.
  Schmidt.
\newblock {A Platform-Independent Component Modeling Language for Distributed
  Real-Time and Embedded Systems}.
\newblock In {\em RTAS '05: Proceedings of the 11th IEEE Real Time on Embedded
  Technology and Applications Symposium}, pages 190--199, Washington, DC, USA,
  2005. IEEE Computer Society.

\bibitem{Blair:01a}
L.~Blair, G.~S. Blair, A.~Anderson, and T.~Jones.
\newblock {Formal Support For Dynamic QoS Management in the Development of Open
  Component-based Distributed Systems}.
\newblock {\em IEEE Software}, 148(3), Nov. 2001.

\bibitem{Czarnecki:06}
K.~Czarnecki and S.~Helsen.
\newblock {Feature-based Survey of Model Transformation Approaches}.
\newblock {\em IBM Syst. J.}, 45(3):621--645, 2006.

\bibitem{Schmidt:06r}
G.~Deng, C.~Gill, D.~C. Schmidt, and N.~Wang.
\newblock {QoS-enabled Component Middleware for Distributed Real-Time and
  Embedded Systems}.
\newblock In I.~Lee, J.~Leung, and S.~Son, editors, {\em Handbook of Real-Time
  and Embedded Systems}. CRC Press, 2007.

\bibitem{Hatcliff:03}
J.~Hatcliff, W.~Deng, M.~Dwyer, G.~Jung, and V.~Prasad.
\newblock {Cadena: An Integrated Development, Analysis, and Verification
  Environment for Component-based Systems}.
\newblock In {\em Proceedings of the 25th International Conference on Software
  Engineering}, pages 160--172, Portland, OR, May 2003.

\bibitem{Kruger:05}
{Jaswinder Ahluwalia and Ingolf H. Krüger and Walter Phillips and Michael
  Meisinger}.
\newblock {Model-Based Run-Time Monitoring of End-to-End Deadlines}.
\newblock In {\em Proceedings of the Fifth ACM International Conference On
  Embedded Software}, pages 100--109, Jersey City, NJ, Sept. 2005. ACM.

\bibitem{Karsai:03}
G.~Karsai, A.~Agrawal, F.~Shi, and J.~Sprinkle.
\newblock {On the Use of Graph Transformation in the Formal Specification of
  Model Interpreters}.
\newblock {\em Journal of Universal Computer Science}, 9(11):1296--1321, 2003.
\newblock \url{www.jucs.org/jucs_9_11/on_the_use_of}.

\bibitem{Schmidt:06x}
A.~Kavimandan, K.~Balasubramanian, N.~Shankaran, A.~Gokhale, and D.~C. Schmidt.
\newblock Quicker: A model-driven qos mapping tool for qos-enabled component
  middleware.
\newblock In {\em ISORC '07: Proceedings of the 10th IEEE International
  Symposium on Object and Component-Oriented Real-Time Distributed Computing},
  pages 62--70, Washington, DC, USA, 2007. IEEE Computer Society.

\bibitem{Gokhale_GTQMAP_RTAS:08}
A.~Kavimandan and A.~Gokhale.
\newblock {Automated Middleware QoS Configuration Techniques using Model
  Transformations}.
\newblock In {\em Proceedings of the $14^{th}$ IEEE Real-Time and Embedded
  Technology and Applications Symposium (RTAS 2008)}, pages 93--102, St. Louis,
  MO, USA, Apr. 2008.

\bibitem{MDABook:03}
A.~Kleppe, J.~Warmer, and W.~Bast.
\newblock {\em MDA Explained: The Model Driven Architecture$(MDA^{TM})$:
  Practice and Promise}.
\newblock Addison-Wesley Longman Publishing Co., Inc., Boston, MA, USA, Apr
  2003.

\bibitem{AIRES:03}
S.~Kodase, S.~Wang, Z.~Gu, and K.~G. Shin.
\newblock {Improving Scalability of Task Allocation and Scheduling in Large
  Distributed Real-time Systems using Shared Buffers}.
\newblock In {\em {Proceedings of the 9th Real-time/Embedded Technology and
  Applications Symposium (RTAS 2003)}}, Washington, DC, May 2003. IEEE.

\bibitem{Schmidt:04y}
A.~S. Krishna, E.~Turkay, A.~Gokhale, and D.~C. Schmidt.
\newblock {Model-Driven Techniques for Evaluating the QoS of Middleware
  Configurations for DRE Systems}.
\newblock In {\em Proceedings of the 11th Real-time Technology and Application
  Symposium (RTAS '05)}, pages 180--189, San Francisco, CA, Mar. 2005. IEEE.

\bibitem{Schmidt:06l}
P.~Lardieri, J.~Balasubramanian, D.~C. Schmidt, G.~Thaker, A.~Gokhale, and
  T.~Damiano.
\newblock {A Multi-layered Resource Management Framework for Dynamic Resource
  Management in Enterprise DRE Systems}.
\newblock {\em Journal of Systems and Software: Special Issue on Dynamic
  Resource Management in Distributed Real-time Systems}, 80(7):984--996, July
  2007.

\bibitem{GME:01}
{\'A}.~L\'{e}deczi, {\'A}.~Bakay, M.~Mar{\'o}ti, P.~V{\"o}lgyesi, G.~Nordstrom,
  J.~Sprinkle, and G.~Karsai.
\newblock {Composing Domain-Specific Design Environments}.
\newblock {\em Computer}, 34(11):44--51, 2001.

\bibitem{WhenHow_DSL_Mernik:05}
M.~Mernik, J.~Heering, and A.~M. Sloane.
\newblock {When and How to Develop Domain-specific Languages}.
\newblock {\em ACM Computing Surveys}, 37(4):316--344, 2005.

\bibitem{Karsai_ENTCS:08}
A.~Narayanan and G.~Karsai.
\newblock Towards verifying model transformations.
\newblock {\em Electronic Notes on Theoretical Computer Science}, 211:191--200,
  2008.

\bibitem{Karsai_ECEASST:08}
A.~Narayanan and G.~Karsai.
\newblock Verifying model transformations by structural correspondence.
\newblock {\em Electronic Communications of the EASST}, 10, 2008.

\bibitem{Neema:02a}
S.~Neema, T.~Bapty, J.~Gray, and A.~Gokhale.
\newblock {Generators for Synthesis of QoS Adaptation in Distributed Real-time
  Embedded Systems}.
\newblock In {\em Proceedings of the ACM SIGPLAN/SIGSOFT Conference on
  Generative Programming and Component Engineering (GPCE'02)}, pages 236--251,
  Pittsburgh, PA, Oct. 2002.

\bibitem{LWCCM}
{Object Management Group}.
\newblock {\em {Lightweight CCM RFP}}, {realtime/02-11-27} edition, Nov. 2002.

\bibitem{RTCORBA:05}
{Object Management Group}.
\newblock {\em {Real-time CORBA Specification}}, {1.2} edition, Jan. 2005.

\bibitem{Ritter:03}
T.~Ritter, M.~Born, T.~Untersch\"{u}tz, and T.~Weis.
\newblock {A QoS Metamodel and its Realization in a CORBA Component
  Infrastructure}.
\newblock In {\em {Proceedings of the $36^{th}$ Hawaii International Conference
  on System Sciences (HICSS'03)}}, page 318, Honolulu, HI, Jan. 2003.

\bibitem{Bogor:03}
Robby, M.~Dwyer, and J.~Hatcliff.
\newblock {Bogor: An Extensible and Highly-Modular Model Checking Framework}.
\newblock In {\em Proceedings of the $4^{th}$ Joint Meeting of the European
  Software Engineering Conference and ACM SIGSOFT Symposium on the Foundations
  of Software Engineering (ESEC/FSE 2003)}, Helsinki, Finland, September 2003.
  ACM.

\bibitem{Schmidt:06w}
N.~Shankaran, D.~C. Schmidt, Y.~Chen, X.~Koutsoukous, and C.~Lu.
\newblock {The Design and Performance of Configurable Component Middleware for
  End-to-End Adaptation of Distributed Real-time Embedded Systems}.
\newblock In {\em Proc. of the 10th IEEE International Symposium on
  Object/Component/Service-oriented Real-time Distributed Computing (ISORC
  2007)}, Santorini Island, Greece, May 2007.

\bibitem{Sharp:98}
D.~C. Sharp.
\newblock {Reducing Avionics Software Cost Through Component Based Product Line
  Development}.
\newblock In {\em Software Product Lines: Experience and Research Directions},
  volume 576, pages 353--370, Aug 2000.

\bibitem{VEST:03}
J.~A. Stankovic, R.~Zhu, R.~Poornalingam, C.~Lu, Z.~Yu, M.~Humphrey, and
  B.~Ellis.
\newblock {VEST: An Aspect-Based Composition Tool for Real-Time Systems}.
\newblock In {\em RTAS '03: Proceedings of the The 9th IEEE Real-Time and
  Embedded Technology and Applications Symposium}, pages 58--69, Toronto,
  Canada, 2003. IEEE Computer Society.

\bibitem{Gorton:07}
L.~Zhu, N.~B. Bui, Y.~Liu, and I.~Gorton.
\newblock {MDABench: Customized benchmark generation using MDA}.
\newblock {\em Journal of Systems and Software}, 80(2):265--282, Feb. 2007.

\bibitem{Quo:97}
J.~A. Zinky, D.~E. Bakken, and R.~Schantz.
\newblock {Architectural Support for Quality of Service for CORBA Objects}.
\newblock {\em Theory and Practice of Object Systems}, 3(1):1--20, 1997.

\end{thebibliography}

% $Id: intro.tex 19213 2009-08-17 10:06:43Z gokhale $
\section{Introduction}
\label{sec:intro}

Contemporary component middleware technologies, such as Enterprise Java
Beans (EJB) and CORBA Component Model (CCM), help to decouple the
development of application logic from the provisioning of non functional
properties of the system, such as domain-specific quality of service
(QoS) policies, which are key to the correct functioning of
distributed real-time and embedded (DRE) systems.  Examples of DRE
system QoS properties that can be configured in the middleware
includes different types of concurrency, handling multiple levels of
priorities of system tasks, publish/subscribe event-driven
communication mechanisms, reliability, security, and multiple
scheduling algorithms, among others.  

Although component middleware separates the QoS configuration complexity
from the application logic, the need to support a variety of
domain-specific QoS policies has made the middleware itself very
complex, which in turn makes the middleware QoS configuration activity
very complex.  Middleware QoS configuration thus involves mapping the
system-level \emph{QoS policies}---which are dictated by domain
requirements---onto the solution space comprising the \emph{QoS
  mechanisms} for tuning the underlying middleware.    

Examples of domain-level QoS policies include (1) the degree of
concurrency required to provide a service, (2) the priorities at which
the different components should run, (3) the alternate protocols that
can be used to request a service, (4) the granularity of sharing among
the application components of the underlying resources, such as
transport level connections, (5) the number and size of outstanding
requests that are permissible at any instant in time, and (6) the
maximum and minimum amount of time to wait for completion of requests,
among many others.  All these domain-specific QoS policies must
eventually be mapped onto the right middleware QoS configurations, which
includes choosing the right set of middleware configuration parameters
and assigning the right values to them. 

An additional dimension of complexity stems from the need to determine
these QoS configurations at different time scales.  This includes
\emph{statically}, \emph{e.g.}, directly hard coded into the
application or middleware; \emph{semi-statically}, \emph{e.g.},
configured at deployment time using metadata descriptors; and/or
\emph{dynamically}, \emph{e.g.}, by modifying QoS configurations at
runtime.   

The complexity involved in the middleware QoS configuration process
raises the following questions: 
\begin{BulletList}
\item How can the domain-specific QoS policies of the system be
 mapped onto QoS configurations of the underlying middleware,
 particularly by DRE system developers who are domain experts but seldom
 possess detailed knowledge of the middleware and its configurability?

\item How are the right set of configuration parameters determined and
  how are valid values for the selected set of QoS configuration
  parameters determined?  Are there any \emph{patterns of usage,}
  \emph{i.e.}, best practices in configuration that can be used across
  a variety of application domains?

\item How can dependency relationships between configuration parameters
  be resolved while ensuring that their interactions do not 
  adversely impact QoS?  These issues arise at the individual component
  level (local) as well as at aggregate intermediate levels, such as
  component assemblies, all the way through the entire application
  (global), and all of these at different time scales, \emph{i.e.},
  design-, deployment- and run-time. 

\end{BulletList}

Without scientific techniques and effective tools to address these
questions, existing approaches to middleware QoS configurations will
often result in QoS mis-configurations that are hard to analyze and
debug.  As a result, failures will stem from a new class of
configuration errors rather than (just) traditional
design/\-implementation errors or resource failures. 

\noindent\textbf{Solution approach $\rightarrow$ Model-to-model
  Transformations:} We address these questions by presenting a technique
based on model-to-model transformations, which is realized in the
context of a tool-chain called \emph{QUality of service pICKER}
(QUICKER)~\cite{Schmidt:06x}.  QUICKER is designed to bridge the gap
shown in Figure~\ref{fig:QoSConfiguration} between: 
\begin{figure}[htb]
\centering
\includegraphics[width=0.5\textwidth,keepaspectratio]{QoSConfiguration}
  \caption{QoS mapping landscape}
  \label{fig:QoSConfiguration}
\end{figure}
\begin{BulletList}
\item \textbf{Functional specification and analysis tools}, such as PICML
  and Cadena~\cite{Hatcliff:03}, that allow specification and analysis of
  application structure and behavior,
\item \textbf{Schedulability analysis tools}, such as
  TIMES~\cite{TIMES:03}, AIRES~\cite{AIRES:03}, VEST~\cite{VEST:03}, that
  perform schedulability and timing analysis to determine the exact
  priorities and time periods for application components, and
\item \textbf{Dynamic QoS adaptation frameworks}, such as the Resource
  Adaptation and Control Engine (RA\-CE)~\cite{Schmidt:06w} and
  QuO~\cite{Quo:97}, that allocate resources to application components,
  monitor the QoS of the system continuously, and apply corrective control
  to modify the QoS configuration of the middleware at runtime.
\end{BulletList}

QUICKER makes the following contributions in addressing the questions
raised above:

\begin{BulletList}
\item QUICKER provides domain-specific modeling languages
      (DSMLs)~\cite{WhenHow_DSL_Mernik:05} that 
      enable DRE system developers to use intuitive abstractions to
      model system-level QoS policies.

\item QUICKER provides a model-to-model transformation process to
      automate the mapping of domain-specific QoS policies expressed
      in the DSML into middleware-specific QoS configurations.  This
      transformation process is verified for correctness using
      structural correspondence~\cite{Karsai_ECEASST:08}. 

\item Dependencies among the configuration options and the correctness
      of the generated configurations at all time scales is resolved
      using automated model checking provided by the
      Bogor~\cite{Bogor:03} model-checking framework. 

\end{BulletList}

% To make the discussion concrete, we present the scientific principles
% behind the model-to-model transformations in QUICKER in the context of
% two domain-specific modeling languages (DSMLs).

The remainder of this paper is organized as follows:
Section~\ref{sec:qos-challenges} describes the key challenges in QoS
configuration of component middleware; Section~\ref{sec:transform}
describes the QUICKER model-to-model transformation approach;
Section~\ref{sec:correctness} discusses how we have employed structural
correspondence and model-checking techniques to verify the correctness
of the QoS configuration process; 
% Section~\ref{sec:eval} provides empirical validation of the generated
% QoS configurations for a representative DRE system; 
Section~\ref{sec:related} compares QUICKER
with related work on model-driven QoS config/-uration/\-adapta/-tion;
and Section~\ref{sec:conclusion} presents concluding remarks.
